Author's response to reviews

Title: A factorial randomized controlled trial to evaluate the effect of micronutrients supplementation and regular aerobic exercise on maternal endothelium-dependent vasodilatation and oxidative stress of the newborn: a study protocol

Authors:

Ana C Aguilar de Plata (caplataa@yahoo.es)
Julio C Mateus (jcmateus@fundacionfes.org)
Robinson Ramirez-Velez (robin640@hotmail.com)
Isabella Echeverry (isabella_ech@hotmail.com)
Jose G Ortega (guileort@yahoo.com)
Mildrey Mosquera (mildrey@yahoo.com)
Blanca Salazar (bsblasalaz@gmail.com)
Sandra L Girón (sandragiron@fundacionfes.org)
Wilmar Saldarriaga (wsaldarriaga0608@yahoo.com)
Miryam Romero (miryamrom@gmail.com)

Version: 6 Date: 26 January 2011

Author’s response to reviews:

Santiago de Cali, January 26 2011

Editors
Trials
RE: MS: 1488023423382489

Dear Editor,

We are sending the revised version of the manuscript “A factorial randomized controlled trial to evaluate the effect of micronutrients supplementation and regular aerobic exercise on maternal endothelium-dependent vasodilatation and oxidative stress of the newborn”. In accordance with the instructions received, we have made all the format changes indicated.

We would like to thank you for accepting our manuscript

Att

Ana C Aguilar de Plata, Julio C Mateus, Robinson Ramirez-Velez, Isabella Echeverry, Jose G Ortega, Mildrey Mosquera, Blanca Salazar, Sandra L Girón, Wilmar Saldarriaga and Miryam Romero